



Citation: Wagenaar I, Post E, Brandsma W, Bowers B, Alam K, Shetty V, et al. (2017) Effectiveness of 32 versus 20 weeks of prednisolone in leprosy patients with recent nerve function impairment: A randomized controlled trial. PLoS Negl Trop Dis 11(10): e0005952. https://doi. org/10.1371/journal.pntd.0005952

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, UNITED STATES

Received: May 2, 2017

Accepted: September 12, 2017

Published: October 4, 2017

Copyright: © 2017 Wagenaar et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting information files.

Funding: The study was funded by The American Leprosy Mission (http://www.leprosy.org), the German Leprosy and TB Relief Association (http:// www.dahw.de), the Netherlands Leprosy Relief (http://leprosyrelief.org), the Ordre de Malte (http:// www.ordredemaltefrance.org), and the Turing Foundation (http://www.turingfoundation.org). EP RESEARCH ARTICLE

Effectiveness of 32 versus 20 weeks of prednisolone in leprosy patients with recent nerve function impairment: A randomized controlled trial

Inge Wagenaar¹*, Erik Post², Wim Brandsma³, Bob Bowers^{4,5}, Khorshed Alam⁴, Vanaja Shetty⁶, Vivek Pai⁷, Sajid Husain⁸, Cita Rosita Sigit Prakoeswa⁹, Linda Astari⁹, Deanna Hagge¹⁰, Mahesh Shah¹⁰, Kapil Neupane¹⁰, Krishna Bahadur Tamang¹¹, The TENLEP study group¹, Peter Nicholls¹², Jan Hendrik Richardus¹

 Dept. of Public Health, Erasmus MC, University Medical Center Rotterdam, Rotterdam, the Netherlands,
KIT Health, Royal Tropical Institute, Amsterdam, the Netherlands,
Independent Leprosy Consultant, Royal Tropical Institute, Amsterdam, the Netherlands,
Independent Leprosy Mission International Bangladesh, Nilphamari, Bangladesh,
Menzies Health Program, The Leprosy Mission International Bangladesh, Nilphamari, Bangladesh,
Menzies Health Institute Queensland, Griffith University, Brisbane, Australia,
Foundation for Medical Research, Mumbai, India,
Bombay Leprosy Project, Mumbai, India,
JALMA institute of Leprosy & Other Mycobacterial Diseases, Agra, India,
Dermatovenereology Dept., Dr Soetomo Hospital–Faculty of Medicine, Universitas Airlangga, Surabaya, Indonesia,
Anandaban Hospital and Mycobacterial Research Laboratories, The Leprosy Mission Nepal, Kathmandu, Nepal,
Lalgadh Leprosy Hospital and Services Centre Dhanusha, Lalgadh, Nepal,
Faculty of Health Sciences, University of Southampton, Southampton, United Kingdom

 \P The members of The TENLEP study group can be found in the "Members of The TENLEP study group" section below.

* inge.wagenaar10@gmail.com

Abstract

Background

While prednisolone is commonly used to treat recent nerve function impairment (NFI) in leprosy patients, the optimal treatment duration has not yet been established. In this "Treatment of Early Neuropathy in Leprosy" (TENLEP) trial, we evaluated whether a 32-week prednisolone course is more effective than a 20-week course in restoring and improving nerve function.

Methods

In this multi-centre, triple-blind, randomized controlled trial, leprosy patients who had recently developed clinical NFI (<6 months) were allocated to a prednisolone treatment regimen of either 20 weeks or 32 weeks. Prednisolone was started at either 45 or 60 mg/day, depending on the patient's body weight, and was then tapered. Throughout follow up, NFI was assessed by voluntary muscle testing and monofilament testing. The primary outcome was the proportion of patients with improved or restored nerve function at week 78. As secondary outcomes, we analysed improvements between baseline and week 78 on the Reaction Severity Scale, the SALSA Scale and the Participation Scale. Serious Adverse Events and the need for additional prednisolone treatment were monitored and reported.

applied for and received the funding. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Results

We included 868 patients in the study, 429 in the 20-week arm and 439 in the 32-week arm. At 78 weeks, the proportion of patients with improved or restored nerve function did not differ significantly between the groups: 78.1% in the 20-week arm and 77.5% in the 32-week arm (p = 0.821). Nor were there any differences in secondary outcomes, except for a significant higher proportion of Serious Adverse Events in the longer treatment arm.

Conclusion

In our study, a 20-week course of prednisolone was as effective as a 32-week course in improving and restoring recent clinical NFI in leprosy patients. Twenty weeks is therefore the preferred initial treatment duration for leprosy neuropathy, after which likely only a minority of patients require further individualized treatment.

Author summary

Nerve damage is a common and severe consequence of leprosy, and it can lead to impairment of sensory (sensibility) or motor (muscle) function of hand, feet and eyes. To prevent that the nerve damage becomes permanent, it is essential that new damage is treated in an early stage. Prednisolone, an immune suppressor, is the drug of choice. However, the ideal treatment duration is unknown. In our study, we compared a prednisolone treatment of 20 weeks with a longer treatment of 32 weeks. After the follow up period of 1.5 years, the percentage of patients with an improved or completely restored nerve function was the same in the two treatment groups, namely 78%. We can conclude that the longer treatment duration was not more effective, and that 20 weeks of prednisolone is sufficient to treat early nerve damage in most patients. We also found that in 15% of the patients neither one of the treatments was effective. For this specific group, more research needs to be done to find if alternative medication or perhaps even longer prednisolone treatment can be helpful.

Introduction

Leprosy is an infectious disease caused by *Mycobacterium leprae*. Since the introduction of antibiotic multidrug treatment (MDT) in the 1980's, the number of leprosy diagnoses has decreased dramatically, and the disease was even declared eliminated as a public health problem at a global level in the year 2000 –i.e. less than 1 case per 10 000 inhabitants. Nevertheless, in 2015 a total of 210 000 new leprosy patients were diagnosed worldwide [1].

A main complication of leprosy is neuropathy, which often causes sensory and motor nerve function impairment (NFI). Untreated NFI can result in deformities of the hands and feet and may also affect the eyes. NFI can develop before MDT has started, but it can also arise during MDT and even several years after leprosy treatment has been completed [2,3]. The risk of developing new NFI within two years of starting MDT can be as high as 65% [4].

To prevent disabilities and deformities in leprosy patients, it is very important to detect and treat neuropathy in an early stage. Neuropathy is commonly treated with prednisolone [5], an immune suppressor that reduces the body's immune responses towards *M. leprae* and relieves the pressure on the nerves by reducing inflammation and oedema [6,7].

Although ideally prednisolone therapy is adjusted to individual needs and response, this is not always feasible in field clinics, which often lack the treatment expertise of referral centres [8]. In these situations, the WHO recommends a standardized prednisolone treatment for 12 weeks [9].

Even though observational studies suggest that prednisolone can improve nerve function in 60–70% of nerves [2, 10–12], this effect has never been established in randomized controlled trials (RCT) [13]. There are indications, however, that a longer treatment duration may be more effective than the WHO standardized treatment. In an RCT in India, type 1 reaction (T1R) patients on a 20-week prednisolone course required less additional prednisolone than patients on a 12-week course [14]. Further research is needed to establish the optimal prednisolone regimen specifically for leprosy patients with NFI.

For this reason, we designed a study entitled "Treatment of Early Neuropathy in Leprosy" (TENLEP), comprising two RCTs aimed at determining how prednisolone treatment best can prevent permanent NFI. In this paper we describe the results of the Clinical trial, in which we evaluated whether a 32-week prednisolone course is more effective than a 20-week course in restoring and improving recent clinical neuropathy (<6 months) [15].

Methods

The TENLEP study was a multicentre, triple blind parallel-group clinical trial, conducted in six referral centres for leprosy in India, Nepal, Bangladesh and Indonesia. A more detailed description of the TENLEP study can be found in the study protocol paper [15].

Ethics statement

Before trial initiation, approvals were obtained in each country from the appropriate ethical review committees: in India from the Indian Council of Medical Research and the Ethics Committee of the Foundation for Medical Research, Mumbai (FMR/IEC/LEP.01b/2011); in Nepal from the Nepal Health Research Council (Reg.no 14/2011); in Indonesia from the Komite Etik Penelitian Kesehatan RSUD Dr. Soetomo Surabaya (100/Panke.KKE/V/2011); in Bangladesh from the Bangladesh Medical Research Council- National Research Ethics Committee (BMRC/NREC/2010-2013/533). Written informed consent was taken from all patients, or their parents in case the patient was under the age of 18 years. The trial was registered in the Netherlands Trial Register (NTR2300) and in the Clinical Trials Registry India (CTRI/2011/ 09/002022 and 23).

Patients

Leprosy patients between 15 and 60 years of age with any recent peripheral NFI (less than 6 months onset) were eligible for the trial. Leprosy diagnosis and classification assignment was confirmed with physical exam by an experienced clinician, physiotherapist exam for neuropathy and slit skin smear microscopy. Ridley-Jopling classification was clinically assigned except in Anandaban hospital, where skin biopsy histopathology was additionally employed. NFI was established with voluntary muscle testing (VMT) and/or monofilament testing (MFT). Patients were excluded if they were pregnant, already receiving prednisolone treatment, suffered from other conditions that may affect the peripheral nervous system, or presented with a single skin lesion on the trunk as the only sign of leprosy. The sample size was calculated to be able to detect 'restored or improved' nerve function in 70% of the intervention group, compared to an assumed proportion of 60% in the control group. This one-tailed hypothesis, using 80% power, 5% significance and allowing for 20% loss to follow-up, lead to a sample size of 720 patients.

Treatment

Prednisolone dose started at 45 mg/day for patients with low weight (\leq 50 kg) and at 60 mg/ day for patients with high weight (>50 kg). The prednisolone dose was then slowly tapered during the treatment period, maintaining a plateau of 20–35 mg/day for 20 weeks in the 32-week arm–depending on weight group. Total dosage and dose over time are previously described [15]. To check the chemical composition of the prednisolone and placebo tablets, a random selection of packages from both treatment groups was evaluated at the start of the trial by the manufacturer (Rubicon), and by an independent Indian laboratory (Medibios Laboratories). After 20 months, the composition was checked again by the Royal Dutch Society of Pharmacists (KNMP). Treatment adherence was checked every month either verbally or by checking the medication package of the previous month.

Randomization and blinding

Patients were randomly allocated to either 20 or 32 weeks of oral prednisolone, using a separate computer-generated blocked randomization sequence for each centre and weight group, blinded within pre-set patient numbered packaging. Patients were kept blinded, as the tablets appeared the same and treatment duration was kept equal by using placebo tablets. In addition, research staff and the statistician were kept blinded until all data analyses were performed. The key to treatment allocation was only broken if a patient had a serious adverse event (SAE) or required individualized treatment for a reaction or worsening NFI.

Assessments

In each centre, monofilament testing and voluntary muscle testing were carried out by two trained assessors, except in the Indonesian centre where ten assessors have performed the tests. Assessments at baseline generally showed good inter-tester reliability [16]. For every patient, the sensory function of six nerves and the motor function of seven nerves were assessed on both left and right body side (total of 26 assessments). Sensory function was tested on three test-sites for each nerve, using a standard set of Semmes-Weinstein monofilaments, with the 200mg filament representing the normal threshold for the hand, and the 2g filament for the foot. For motor function assessment, the 0-5 Medical Research Council (MRC) scale was used [17]. The exact test methods and sites are previously described [13]. When the total monofilament score for a nerve was 3 or more, the sensory nerve function was considered impaired. A motor nerve scoring less than 5 on the Medical Research Council scale was also regarded as impaired. Follow-up assessments for VMT and MFT were carried out monthly during the treatment period (up to week 32), and at week 52 and 78. At baseline and the end of the study a Screening of Activity Limitation and Safety Awareness (SALSA) scale [18] and a Participation (P) scale [19] were completed for each patient. In addition, reaction severity was monitored with a Reaction Severity Scale (RSS) [20] at baseline, week 32, 52 and 78. When a patient did not show up for their follow-up appointment, a telephone call or in some cases a home visit was made in an effort to get the patient visit the clinic for assessments. When indicated in advance, a patient was allowed to miss one assessment, and the medication was provided for eight weeks.

As previously described, if nerve function deteriorated or new reaction symptoms developed, the trial treatment was stopped and an individually modified treatment was provided [15]. A decision tree was used to help clinicians decide whether or not patients should continue in the trial or receive individualized treatment in case of new NFI or reaction symptoms. Once individualized treatment was warranted, it was up to the clinician to decide on further patient management. The definition used to determine deteriorating MFT was: an increase of

Table 1. Overview of primary and secondary outcomes.

	Outcome	Definitions
Primary outcome	Proportion of patients with restored or improved nerve function	Using a composite score. <u>Restored</u> : all nerves back to normal function <u>Improved</u> : more restored* and/or improved** nerves than unchanged and/or deteriorated*** nerves <u>Unchanged</u> : same number of nerves impaired <u>Deteriorated</u> : more deteriorated nerves than improved and/or restored nerves
Secondary outcomes	1. Proportion of restored, improved, unchanged, deteriorated and fully impaired nerves for each specific nerve—e.g. ulnar nerve	For six sensory nerves and seven motor nerves
	2. Count of impairments	Per patient 26 nerves were assessed. Each impaired nerve adds 1 point to the total score
	3. Improvement Reaction Severity Scale	Improvement: reduction in RSS of 3 or more points on the total score, or a reduction of 2 or more points on any individual item of the scale
	4. Improvement in SALSA-scale and P-scale	Improvement: classified in a better category (extreme, severe, moderate, mild, none)
	5. Proportion of patients with Serious Adverse Events	
	6. Proportion of patients needing additional steroids for treatment of reaction or worsening NFI	

* restored: back to normal function

** improved: decrease of 3 points for MFT or increase of 1 point for VMT

***: deteriorated: increase of 3 points for MFT or decrease 1 point for VMT

https://doi.org/10.1371/journal.pntd.0005952.t001

6 points or more on the score per nerve since the last assessment, or an increase of 3 points or more on the score per nerve that was confirmed on the consecutive visit. VMT deterioration was defined as: a reduction in VMT score by two or more points or a reduction of 1 point on two consecutive visits. Removal from the trial treatment due to reactions was based on new or deteriorating reaction symptoms.

Patients who developed a serious adverse event (SAE) were removed from trial treatment and provided with individualized care. For both deteriorating and SAE cases, data collection continued up to the full 78 weeks.

Primary and secondary outcomes

The primary study outcome was the proportion of patients with restored or improved nerve function (of all nerves) as measured by MFT and/or VMT at 78 weeks. Secondary outcomes considered six variables: individual nerves, impairment counts, RSS, SALSA-scale and P-scale scores between baseline and week 78, SAE's between the intervention and control groups, and the proportion of patients needing additional prednisolone with considering differences in timing, dose and duration. Definitions for restoration, improvement and deterioration in these respective categories for secondary and primary outcomes were as previously described and are depicted in Table 1 [15].

Data analysis

Data were entered at each centre in an Access Database and then combined to be analysed in Stata. Data were analysed according to the modified intention-to-treat principle: data of all randomized patients who matched the inclusion criteria were analysed for week 78, whether they had finished treatment or not, including all patients lost to follow-up. To handle missing data of patients lost to follow-up, the last observation carried forward method was used. For

patients who received additional prednisolone, the assessment recorded at the time when additional prednisolone was first prescribed was carried forward. Only nerves with new impairments (<6 months) were included in the analyses. The primary and secondary outcomes were analysed using a Chi-Square test. A difference between treatment groups was considered significant when the p-value was < 0.05.

Results

Participants

A total of 875 leprosy patients were enrolled in the trial, of whom 432 were randomized to the 20-week arm and 443 to the 32-week arm. Patients were recruited between February 2012 and October 2013, and the last follow-up data were collected in July 2015. The flow diagram in Fig 1 illustrates the number of patients followed up and the reasons for drop out. Seven of the



Fig 1. Flow diagram of number of patients enrolled, randomized, treated and followed-up.

https://doi.org/10.1371/journal.pntd.0005952.g001

randomized patients were excluded in the final analyses: one patient who had missing baseline MFT assessments and six patients who did not meet the inclusion criteria of having recent NFI. The number included in the analyses reported here was therefore 868.

Baseline data

The intake and distribution of several patient characteristics per research centre are presented in Table 2. Table 3 shows the baseline demographic and clinical characteristics of the total patient group. Differences in demographic and clinical characteristics did not reach statistical significance between the groups. At baseline, sensory function was more often impaired than motor function: for both groups the median number of nerves with impaired sensory function was 3, ranging from 0–12, while the median number of nerves with impaired motor function was 1 (0–13). The proportions of sensory and motor impairment per nerve are shown in Fig 2.

Number analysed

Of the 868 patients enrolled in the study, the trial treatment period—the first 32 weeks—was completed by 281 (65.5%) in the 20-week arm and 293 (66.7%) in the 32-week arm. Complete follow-up data until week 78 were collected for 230 (82%) of those patients in the 20-week arm and 219 (75%) patients in the 32-week arm (see Fig 1). At week 78, follow-up data were collected for 229 additional patients who did not complete trial treatment due to new NFI, new or recurrent reactions, SAEs or loss to follow-up. For the intention-to-treat analyses, the primary and secondary outcomes were analysed using the data of all patients who met the inclusion criteria (n = 868). A separate per protocol analysis was carried out including only patients who had completed treatment (n = 574).

Primary outcome

The proportion of patients with restored or improved nerve function at week 78 was almost similar in both groups: 78.1% in the 20-week arm and 77.5% in the 32-week arm (p = 0.821). At week 32 and week 52, this outcome was not significantly different between the two groups either. Fig 3 shows the proportion of patients in each category of improvement. In the 20-week arm, more patients showed completely restored nerve function than in the 32-week arm, 23.5% against 18.7%. The per-protocol analysis, leaving out treatment non-compliers, resulted

Centre	Number of patients	Gender (% female)	Mean age	Low weight group (%)	20-week arm (%)
Anandaban Hospital <i>Kathmandu, Nepal</i>	57	26%	36.7	37%	47%
Foundation for Medical Research <i>Mumbai, India</i>	82	15%	31.8	34%	49%
JALMA institute of Leprosy & Other Mycobacterial Diseases <i>Agra, India</i>	230	17%	31.5	53%	50%
Lalgadh Leprosy Hospital and Services Centre Dhanusha, Nepal	166	28%	34.3	45%	50%
Danish Bangladesh Leprosy Mission Nilphamari, Bangladesh	232	22%	39.1	51%	50%
Dr. Soetomo Hospital Surabaya and Madura, Indonesia	101	25%	31.9	46%	48%

Table 2. Intake per research centre.

https://doi.org/10.1371/journal.pntd.0005952.t002

Table 3. Baseline demographics and clinical characteristics of the 868 patients.

	20-week arm		32-week arm		
	(n = 429)		(n = 439)		
Gender (% female)	101	(23.4%)	90	(20.4%)	
Age (mean ±SD)	34.7	(12.1)	34.3	(12.1)	
Literate (%)	241	(56.5%)	247	(56.3%)	
MB/PB (% MB)	339	(79.2%)	357	(81.0%)	
RJ classification					
TT	9	(2.1%)	20	(4.6%)	
BT	212	(49.4%)	203	(46.2%)	
BB	61	(14.2%)	55	(12.5%)	
BL	67	(15.6%)	70	(16.0%)	
LL	40	(9.3%)	51	(11.6%)	
PN	40	(9.3%)	40	(9.1%)	
Average smear (mean ±SD)					
ТТ	0	(0)	0	(0)	
ВТ	0.3	(0.8)	0.2	(0.8)	
BB	0.3	(0.7)	0.6	(1.2)	
BL	2.0	(1.5)	2.0	(1.7)	
LL	3.0	(1.8)	3.2	(1.7)	
PN	0.4	(0.8)	0	(0)	
Disability grade					
0	171	(39.9%)	165	(37.6%)	
1	158	(36.8%)	176	(40.1%)	
2	100	(23.3%)	98	(22.3%)	
T1R					
Mild	66	(15.4%)	59	(13.4%)	
Severe	14	(3.3%)	24	(5.5%)	
T2R					
Mild	11	(2.6%)	14	(3.2%)	
Severe	2	(0.5%)	6	(1.4%)	

MB: Multibacillary; PB: Paucibacillary; RJ: Ridley-Jopling; T1R: Type 1 Reaction; T2R: Type 2 Reaction

https://doi.org/10.1371/journal.pntd.0005952.t003

in an overall slightly higher proportion of patients with restored and improved nerves. However, again no significant difference was found between the groups: 81.9% in the 20-week arm and 81.7% in the 32-week arm (p = 0.960).

Secondary outcomes

Table 4 presents the outcomes per nerve, for six sensory and seven motor nerves. The proportion of restored and improved motor nerves was overall higher than for sensory nerves. The sensory function of the radial, sural and posterior tibial nerves had the highest proportion of improvement and restoration. The motor function of the ulnar and posterior tibial nerves improved most between baseline and week 78. The median count of impairments at baseline and at week 78 is shown in Table 5. 75.7% of the patients in the 20-week arm showed an improvement in count of impairment, compared to 70.8% in the 32-week arm (p = 0.149). Table 5 also presents the results of the Reaction Severity scale, the SALSA scale and the P-scale for every assessment point. The proportion of patients with improved in RSS score at week 78 was higher in the 32-week arm (70%) than in the 20-week arm (65%), but this difference was





https://doi.org/10.1371/journal.pntd.0005952.g002

not statistically significant (p = 0.09). At 78 weeks, the day-to-day situation for patients had barely improved: the scores of both SALSA and P-scale reduced slightly between baseline and week 78. The change over time did not differ significantly between the two arms (p = 0.638 for SALSA and p = 0.543 for P-scale).



Fig 3. Percentage of patients with restored, improved, unchanged and deteriorated nerve function impairment at week 78 as primary outcome.

https://doi.org/10.1371/journal.pntd.0005952.g003

Table 4. Outcomes per nerve at week 78.

		n	Normal	Restored	Improved	Unchanged	Deteriorated
SENSORY			%	%	%	%	%
Trigeminal	20 weeks	858	97.4	1.5	-	0.9	0.1
	32 weeks	878	98.2	1.4	-	0.1	0.3
Ulnar	20 weeks	830	60.2	17.1	9.3	5.4	8.0
	32 weeks	850	55.3	17.8	11.9	5.7	9.4
Median	20 weeks	839	69.0	14.4	5.7	4.1	6.8
	32 weeks	860	67.1	12.2	8.7	4.0	8.0
Radial	20 weeks	814	49.6	24.1	11.4	6.0	8.9
	32 weeks	845	47.5	21.7	14.1	5.4	11.4
Posterior tibial*	20 weeks	810	51.4	17.0	11.0	9.9	10.7
	32 weeks	839	46.6	16.6	16.2	11.1	9.5
Sural*	20 weeks	784	41.5	14.8	17.6	14.2	12.0
	32 weeks	827	37.1	16.1	17.7	20.9	8.2
MOTOR		n	%	%	%	%	%
Eye closure	20 weeks	852	90.3	4.9	1.8	2.1	0.9
	32 weeks	871	91.9	4.8	1.5	1.0	0.8
Ulnar	20 weeks	826	61.0	15.3	7.1	10.4	6.2
	32 weeks	855	60.9	13.1	8.2	12.4	5.4
Median	20 weeks	850	84.5	8.1	0.6	2.7	4.1
	32 weeks	870	84.6	8.3	1.3	3.2	2.6
Radial	20 weeks	856	95.7	3.0	0.1	0.2	0.9
	32 weeks	877	95.4	2.9	0.1	0.6	1.0
Lateral popliteal (dorsiflexion foot)	20 weeks	850	90.5	5.9	1.1	1.7	0.9
	32 weeks	871	89.8	5.9	0.8	2.3	1.3
Posterior tibial up	20 weeks	833	79.4	10.8	2.2	4.6	3.1
	32 weeks	861	81.2	9.1	1.7	5.7	2.3
Posterior tibial down	20 weeks	826	77.2	11.6	0.4	5.8	5.0
	32 weeks	860	76.5	12.9	0.4	6.9	3.4

*Difference between arms is statistically significant

https://doi.org/10.1371/journal.pntd.0005952.t004

Additional prednisolone, to treat new or deteriorating NFI and reactions, was required in 68 (16%) patients in the 20-week arm and 65 (15%) patients in the 32-week arm. This difference was not statistically significant. Interestingly, the majority (38/68) of these patients in the 20-week arm needed additional prednisolone between week 21 and week 32, while in the 32-week arm additional prednisolone was given mostly between week 33 and 52 (56/65). In the 32-week arm, the additional prednisolone given before week 32 was largely (78%) due to reaction—with or without accompanying worsening of NFI, while the additional prednisolone given after week 32 was more often for deteriorating NFI without reaction (55% of the cases needing additional prednisone did not have other reaction symptoms). Fig 4 shows the time until the first event requiring an additional prednisolone. The dose and duration of additional prednisolone treatment did not significantly differ between the two groups.

When zooming in on the group of patients that needed additional prednisolone (independent of treatment arm), it became apparent that their baseline characteristics differed from the group that did not need additional prednisolone. <u>Table 6</u> gives an overview of these characteristics. Next to the presented variables, patients needing additional prednisolone during or after

	20-week arm			32-week arm		
	n median		n	median	Test statistic	
		(IQ range)		(IQ range)		
Count of impairments						
Baseline sensory	429	3 (1–5)	439	3 (1–6)	NS	
Baseline motor	429	1 (1–3)	439	1 (1–3)	NS	
Total count baseline	429	4 (2–8)	439	4 (2–8)	NS	
Week 78 sensory	338	1 (0-4)	343	2 (0-4)	NS	
Week 78 motor	338	0 (0–2)	343	1 (0–2)	P<0.05	
Total count week 78	338	2 (0–5)	343	2 (1–6)	NS	
Difference total count baseline- week 78	338	2 (0-4)	343	2 (0-4)	NS	
Reaction Severity Scale						
Baseline	426	4.5 (2–8.5)	438	5.5 (2.5–9.5)	NS	
Week 32	425	2 (0–5.5)	437	2.5 (0–6)	NS	
Week 52	427	2 (0–5.5)	437	2.5 (0–6)	NS	
Week 78	422	2 (0–5.5)	432	2 (0–6)	NS	
Difference total count baseline- week 78	420	2.5 (0–5)	432	3.5 (0–6)	NS	
SALSA scale						
Baseline	402	23 (20–35)	412	24 (20–35)	NS	
Week 78	269	20 (20–24)	272	20.5 (20–26)	NS	
Difference baseline- week 78	269	0 (-1–3)	272	0 (-1–3)	NS	
P-Scale						
Baseline	402	12.5 (1–21)	412	9 (0–20)	NS	
Week 78	269	5 (0–18)	273	6 (0–18)	NS	
Difference baseline- week 78	269	0 (-16–7)	273	0 (-18–5)	NS	

Table 5. Secondary outcomes: impairment, reaction severity scale, SALSA scale, P-scale.

IQ: inter-quartile; NS: not significant

https://doi.org/10.1371/journal.pntd.0005952.t005

the trial period also showed significantly more often other reaction signs, i.e. inflamed and raised skin lesions, peripheral oedema, nerve pain and tenderness.

Adverse events

There were seven deaths, three in the 20-week arm and four in the 32-week arm; but none were related to trial treatment. SAE occurred in both groups, but were more often reported in the intervention group: 12 cases (2.7%) in the 32-week arm and four cases (0.9%) in the 20-week arm [21]. This difference was statistically significant (p = 0.04). The main reported SAE's were hypertension (6), diabetes (6), and peptic ulcer (2): all commonly recognized side effects associated with prednisolone treatment. Minor side effects were reported in 66% in the 32-week arm and 68% in the 20-week arm. Detailed results on adverse events will be reported separately.

Additional analyses

We used logistic regression to assess the (univariate) association between clinical and demographic characteristics recorded at baseline and the primary outcome of improved or restored nerve function. The five characteristics that demonstrated a significant correlation with the primary outcome were weight group, body mass index (BMI), gender, WHO disability grade and Eye Hand Foot (EHF) score. A patient with a higher BMI had higher odds on improved or



https://doi.org/10.1371/journal.pntd.0005952.g004

restored nerve function at 78 weeks (OR = 1.09 (1.04-1.14)). The high weight group (>50 kg) had a 1.43 (1.09–1.86) higher probability of a positive primary outcome than the low weight group. Furthermore, women were 1.46 (1.06–2.10) times more likely to have a restored or improved nerve function than men. The subgroup analysis also demonstrated that patients having a grade 1 or 2 impairment at baseline were less likely to have restored or improved nerve function at week 78 (OR Grade 1: 0.73 (0.54–0.98) and OR Grade 2: 0.19 (0.12–0.28)). A

		Additional predn	Additional prednisolone		No additional prednisolone		
		(n = 134)		(n = 734)			
WHO classification	% MB	120	(89.6%)	576	(78.5%)	0.003	
Ridley-Jopling						<0.001	
classification	ТТ	2	(1.5%)	27	(3.7%)		
	ВТ	61	(45.5%)	354	(48.2%)		
	BB	10	(7.5%)	106	(14.4%)		
	BL	32	(23.9%)	105	(14.3%)		
	LL	23	(17.2%)	68	(9.3%)		
	PN	6	(4.5%)	74	(10.1%)		
Skin smear (BI)	Mean	1.4		0.7		<0.001	
	Positive	60	(44.8%)	157	(21.4%)	<0.001	
Eye-hand-foot score	Mean		1.9		1.5	0.014	
	Positive	94	(70.2%)	438	(59.7%)	0.022	
Type 1 reaction	Positive	44	(32.8%)	119	(16.2%)	<0.001	
Type 2 reaction	Positive	10	(7.2%)	22	(3.0%)	0.011	
Weight group	% high weight	82	(61.2%)	375	(51.1%)	0.031	
BMI	Mean	20.4		19.8		0.024	

Table 6. Overview of differences in baseline characteristics between the group that needed additional prednisolone and the group that did not need additional prednisolone.

https://doi.org/10.1371/journal.pntd.0005952.t006

higher EHF score also reduced the odds of a good outcome (OR: 0.73 (0.67–0.80). The following baseline characteristics had no influence on improved nerve function: WHO classification, Ridley-Jopling classification, BI (skin smear) and the presence or absence of severe Type 1 or Type 2 reaction.

Using a Cox Proportional Hazard model, we assessed whether there was a trend across the entire duration of follow up regarding primary outcome -i.e. not limited to 32, 52 and 78 weeks. The conclusion from this analysis was similar to what has been described above, there was no effect of treatment arm on primary outcome.

Discussion

In both treatment arms in our study, a large majority (78%) of patients showed improvement and restoration of nerve function, indicating that 20-week prednisolone treatment is sufficient for most patients. However, around 15% of the patients required individualized further treatment because of reactions or deteriorating NFI. Notably, the proportion of patients needing additional prednisolone was similar in both groups (15% and 16%), and the duration and dose of the additional prednisolone treatment did not differ either. However, we did see a difference in timing of the need for additional prednisolone. New NFI and reactions were primarily reported in the first few weeks after prednisolone treatment had ended, thus after 20 weeks and 32 weeks respectively for the control and intervention arm. A similar rebound effect was observed in the Trials in Prevention of Disability (TRIPOD) study and in the study of Rao et al. [14,22]. The longer prednisolone treatment in our study seems to merely postpone the immune response in some patients.

The roughly 15% of patients that required additional prednisolone differed at baseline from the group that did not need extra prednisolone. This former group had significantly more MB patients, mainly BL and LL, higher average skin smear, higher EHF score, had T1R and T2R more often at baseline and showed more often increased signs of reaction, i.e. inflamed and raised skin lesions, peripheral oedema, nerve pain and tenderness. Further studies of an even longer prednisolone course than assessed in this study might be beneficial for these higher risk patients, as it seems that the immune response against dead *M. leprae* bacilli is too persistent to be permanently suppressed by a 32-week prednisolone course [23,24]. At present, however, our results reinforce that 20 weeks tapered prednisolone with an extended higher dose plateau is sufficient treatment for the majority of recent leprosy neuropathy cases, while extended individualized treatment will be needed for some patients, which may be associated with commonly recognized risk factors for complications.

Women overall had a better response to prednisolone then men. This was unexpected as previous studies show that women generally have longer delay in presenting with leprosy and may then have more severe NFI at time of diagnosis with less chance for recovery [25-27]. Though, in our study we only included patients with a delay of less than 6 months and therefore a pre-selection of women with recent NFI was made.

The group of patients needing additional prednisolone had significantly more often certain clinical baseline characteristics that are frequently associated with reduced outcomes, such as high BI, MB classification and reactions. When analysing the correlation between baseline characteristics and the primary outcome, i.e. improved/recovered vs. unchanged/deteriorated NFI, we had expected that patients with deteriorated NFI would also have significantly more often a high BI, MB classification and reactions at baseline. This was not the case, however. One explanation may be that the group of 'unchanged' patients diluted the group of patients with a deterioration, and made the association between these specific baseline characteristics and primary outcome less strong.

The mechanisms behind nerve damage in leprosy are generally not well understood and challenging to study. Nerve biopsies can give some insight in the pathophysiology of neuropathy, but are limited to a single point in time and do not provide information about the onset or further development of impairment [23]. Several mechanisms are described in literature though. In vitro, demyelination has been shown to occur when M. leprae invades the Schwann cells [28-30]. This process can take place in an early stage, when the immune system has not been activated yet. Secondary immune responses likely play a role via inflammatory cytokines [31–34] and T-cells [35–38], leading to demyelination or damaging of Schwann cells. Last, mechanical effects such as oedema and ischemia [23,39] can lead to axonal loss. Nerve regeneration can occur after the inflammation is controlled; however, the duration of the inflammation and nerve scarring may influence outcome [7, 39]. As a strong immunomodulator, prednisolone can impact secondary responses to reduce the inflammation, reducing oedema and scar formation [7]; thereby, providing time and space for nerve regeneration to occur if applied within <6 months of symptoms [2,8,40, 41]. When the delay in prednisolone treatment is too long (>6 months), the damage to the nerve is considered to be irreversible. From previous studies [5,42], it seems that NFI of shorter onset duration prior to treatment has better chances on improvement.

Observational studies in leprosy neuropathy have shown that prednisolone improves nerve function [2,10-12]. Although prednisolone is generally considered an important drug for improving NFI, it is not known by what mechanisms or to what extent prednisolone is responsible for the reported improvements, such as in placebo controlled randomized controlled trials like the TRIPOD studies [22,43]. Several studies have also indicated that a high proportion of patients (33-75%) experience spontaneous nerve function improvement even if left untreated, depending on the severity of NFI and type of nerve [22,44-46]. Even in 51% of placebo patients with old NFI (> 6 months of symptoms) improvement can be demonstrated without prednisolone treatment [45]. A placebo-controlled RCT could be an important next step to investigate the actual effect of prednisolone on the treatment of recent NFI. Alternatives for prednisolone should continue to be sought, as so far no treatments have been proven effective. For example, azathioprine and cyclosporine, immunosuppressants used to treat reactions, show only limited effect in studies inclusive of but not specifically for NFI [46-48]. Nerve decompression, a surgical method to relieve severe inflammatory pressure, is sometimes used to improve ongoing neuropathy in leprosy; however, controlled randomized trial evidence on the effectiveness of this method is lacking as well [49].

The strongest point of our study is the multi-centre, multi-country design which enabled representative sampling across leprosy demographics in South East Asia, where the bulk of global leprosy cases are diagnosed. This RCT was unique in maintaining a middle plateau for a longer time with a relatively high dose of prednisolone and demonstrated that a 20-week treatment was sufficient for the majority of recent neuropathy cases. This RCT was also able to demonstrate that baseline BMI of the patient was directly related to outcomes. This highlights the practical and clinical relevance for contextual realities, such as extreme poverty and malnutrition present within some leprosy-affected populations [50–52]. Such issues can represent individual complications for health and healing during a neuropathy episode.

At the conclusion of the study, three limitations became apparent. At the 78-week time point, only 72% of the patients in the 20-week arm and 78% of the patients in the 32-week arm provided data, exceeding the Evidence Based Medicine cut off of 20%, leading to a loss of validity [53]. In addition, treatment data was absent at other time points due to SAE, the need for additional prednisolone or temporary loss-to-follow up. To limit potential bias related to nonrandom loss of patients, data was analysed according to the intention to treat principle, using the last observation carried forward approach. By using this approach, the results of this study are likely conservative with the actual improving effect of prednisolone on NFI possibly even larger. A second limitation is that the duration of NFI prior to intake was self-reported, which may have allowed recruitment of patients with previous symptoms beyond the 6 month cutoff for inclusion and who were, therefore, less likely to demonstrate nerve improvement with treatment [5,41]. The third limitation is that Ridley-Jopling classifications were based on clinical aspects only, except for one centre that performed histopathology. The results of the logistic regression to evaluate the effect of RJ classification on the final result should therefore be interpreted with care.

We conclude that a 20-week course is sufficient to improve nerve function in 78% of patients with recent NFI. Future studies should focus on improved regimens or alternative treatments, as approximately 15% of patients with recent NFI require additional prednisolone. Moreover, alternative therapies could potentially reduce risks for developing steroid-dependency and other long term treatment side effects. Secondly, it is important to better unravel the pathophysiology of nerve function impairment and to study the actual effect of prednisolone on immune function as it relates to NFI. Understanding the mechanisms behind NFI could lead to alternative, more effective solutions for the treatment of NFI and the prevention of irreversible impairments and subsequent disabilities.

Supporting information

S1 Data. Dataset TENLEP clinical trial. (XLSX)

Acknowledgments

We are immensely grateful to the patients who were willing to participate in this study.

Members of The TENLEP study group

Kallyan Kumar Kunda¹, Anil Soren¹, Kolpona Roy¹, Gospi Capadia², Ashish Khodke², Kulbushan Pawar², Mishra Nath Kameshwar³, Bipin Kumar³, Hemant Singh³, Bagus Haryo Kusumaputra⁴, Irmadita Citrashanty⁴, Sylvia Anggraeni⁴, Menul Ayu Umborowati⁴, Renata Mayangsari⁴, Riski Haris Sasongko, Agustina⁴, Asmahani Thohiroh⁴, Ram Babu Bista⁵, Pawan Parajuli⁵, Apsara Ghimiri⁵, Shyam Pariyar⁶, Jeevan Thapa⁶ and Shankar Paudyel⁶.

¹ Rural Health Program, The Leprosy Mission International Bangladesh, Nilphamari, Bangladesh

² Foundation for Medical Research, Mumbai, India

³ JALMA institute of Leprosy & Other Mycobacterial Diseases, Agra, India

⁴ Dermatovenereology Dept., Dr Soetomo Hospital–Faculty of Medicine, Universitas Airlangga, Surabaya, Indonesia

⁵ Anandaban Hospital and Mycobacterial Research Laboratories, The Leprosy Mission Nepal, Kathmandu, Nepal

⁶ Lalgadh Leprosy Hospital and Services Centre Dhanusha, Lalgadh, Nepal

Author Contributions

Conceptualization: Inge Wagenaar, Erik Post, Wim Brandsma, Peter Nicholls, Jan Hendrik Richardus.

Data curation: Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Vivek Pai, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Mahesh Shah, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.

Formal analysis: Inge Wagenaar, Erik Post, Wim Brandsma, Peter Nicholls, Jan Hendrik Richardus.

Funding acquisition: Erik Post.

- **Investigation:** Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Vivek Pai, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.
- **Methodology:** Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Vivek Pai, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Mahesh Shah, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.

Project administration: Erik Post, Wim Brandsma, Peter Nicholls, Jan Hendrik Richardus.

Resources: Erik Post.

Software: Peter Nicholls.

Supervision: Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Vivek Pai, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Mahesh Shah, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.

Validation: Peter Nicholls.

Visualization: Peter Nicholls.

- Writing original draft: Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Mahesh Shah, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.
- Writing review & editing: Inge Wagenaar, Erik Post, Wim Brandsma, Bob Bowers, Khorshed Alam, Vanaja Shetty, Sajid Husain, Cita Rosita Sigit Prakoeswa, Linda Astari, Deanna Hagge, Mahesh Shah, Kapil Neupane, Krishna Bahadur Tamang, Peter Nicholls, Jan Hendrik Richardus.

References

- 1. World Health Organization, Weekly Epidemiological Record. 2016. 91(35): p. 405–420.
- Saunderson P., The pattern of leprosy-related neuropathy in the AMFES patients in Ethiopia: definitions, incidence, risk factors and outcome. Leprosy Review, 2000. 71(3): p. 285–308. PMID: <u>11105488</u>
- Rosenberg N.R., Faber W.R., and Vermeulen M., Unexplained delayed nerve impairment in leprosy after treatment. Leprosy Review, 2003. 74(4): p. 357–365. PMID: 14750581
- 4. Croft R.P., et al., A clinical prediction rule for nerve function impairment in leprosy patients-revisited after 5 years of follow-up. Lepr Rev, 2003. 74(1): p. 35–41. PMID: <u>12669931</u>
- Naafs B., Bangkok Workshop on Leprosy Research. Treatment of reactions and nerve damage. International journal of leprosy and other mycobacterial diseases: official organ of the International Leprosy Association, 1996. 64(4 Suppl): p. S21–28.
- Lockwood D.N.J., Steroids in leprosy type 1 (reversal) reactions: Mechanisms of action and effectiveness. Leprosy Review, 2000. 71(SUPPL.): p. S111–S114.
- Naafs B., Treatment of reactions and nerve damage. International Journal of Leprosy, 1996. 64(4 SUPPL.): p. S21–S28.

- Britton W.J., The management of leprosy reversal reactions. Leprosy Review, 1998. 69(3): p. 225–234. PMID: 9805878
- World Health Organization, WHO Expert Committee on Leprosy- Seventh report, in World Health Organization technical report series No. 874. 1998: Geneva.
- Britton W.J. and Lockwood D.N.J., Leprosy. Lancet, 2004. 363(9416): p. 1209–1219. <u>https://doi.org/10.1016/S0140-6736(04)15952-7 PMID: 15081655</u>
- Croft R.P., Richardus J.H., and Smith W.C.S., Field treatment of acute nerve function impairment in leprosy using a standardized corticosteroid regimen—First year's experience with 100 patients. Leprosy Review, 1997. 68(4): p. 316–325. PMID: 9503867
- Schreuder P.A.M., The occurrence of reactions and impairments in leprosy: Experience in the leprosy control program of three provinces in northeastern Thailand, 1978–1995. III. Neural and other impairments. International Journal of Leprosy, 1998. 66(2): p. 170–181.
- van Veen N.H., et al., Corticosteroids for treating nerve damage in leprosy. Cochrane Database Syst Rev, 2007(2): p. CD005491.
- Sundar Rao P.S.S., et al., Multi-centre, double blind, randomized trial of three steroid regimens in the treatment of type-1 reactions in leprosy. Leprosy Review, 2006. 77(1): p. 25–33. PMID: <u>16715687</u>
- Wagenaar I., et al., Two randomized controlled clinical trials to study the effectiveness of prednisolone treatment in preventing and restoring clinical nerve function loss in leprosy: the TENLEP study protocols. BMC Neurology, 2012(1471–2377 (Electronic)).
- Brandsma J.W., et al., Reliability of clinical nerve function assessment in peripheral neuropathies. Leprosy Review, 2014. 85(1): p. 29–35. PMID: 24974440
- 17. O'Brien M.D., Aids to the examination of the peripheral nervous system. The Guarantors of Brain. Vol. 5th edition. 2010: Elsevier Health Sciences.
- The SALSA Collaborative Study Group. Screening Activity Limitation and Safety Awareness Users Manual V1.1. 2010 [cited 2012 September 28th]; Available from: http://www.ilep.org.uk/fileadmin/ uploads/Documents/Infolep_Documents/Salsa/SALSA_manual_v1.1pdf.pdf.
- van Brakel, W.H. Participation Scale Users Manual v. 6.0 2010 [cited 2012 September 28th]; Available from: http://www.ilep.org.uk/fileadmin/uploads/Documents/Infolep_Documents/Participation_Scale/ Participation_Scale_Users_Manual_v._6.0.pdf.
- **20.** Walker S.L., et al., Development and validation of a severity scale for leprosy type 1 reactions. PLoS Neglected Tropical Diseases, 2008. 2(12).
- Post E., Wagenaar I., Brandsma W., Nicholls P., Richardus J.H., Prednisolone Adverse Events in the Treatment of Leprosy Neuropathy. In preparation, 2016.
- van Brakel W.H., et al., The prognostic importance of detecting mild sensory impairment in leprosy: A randomized controlled trial (TRIPOD 2). Leprosy Review, 2003. 74(4): p. 300–310. PMID: 14750575
- Scollard D.M., Truman R.W., and Ebenezer G.J., Mechanisms of nerve injury in leprosy. Clinics in Dermatology, 2015. 33(1): p. 46–54. https://doi.org/10.1016/j.clindermatol.2014.07.008 PMID: 25432810
- 24. Shetty V.P., et al., Persistence of Mycobacterium leprae in the peripheral nerve as compared to the skin of multidrug-treated leprosy patients. Leprosy Review, 1992. 63(4): p. 329–336. PMID: 1479872
- 25. Morrison A., A woman with leprosy is in double jeopardy. Leprosy Review, 2000. Jun; 71(2):128–43 PMID: 10920609
- Sarkar R., and Pradhan S., Leprosy and Women. Int J Womens Dermatol, 2016. Oct 25; 2(4):117–121 https://doi.org/10.1016/j.ijwd.2016.09.001 PMID: 28492023
- John A.S., Sundar Rao P.S., Das S., Assessment of needs and quality care issues of women with leprosy. Leprosy Review, 2010. 81, 34–40 PMID: 20496568
- Oliveira R.B., et al., Expression of toll-like receptor 2 on human schwann cells: A mechanism of nerve damage in leprosy. Infection and Immunity, 2003. 71(3): p. 1427–1433. https://doi.org/10.1128/IAI.71. 3.1427-1433.2003 PMID: 12595460
- Rambukkana A., et al., Contact-dependent demyelination by Mycobacterium leprae in the absence of immune cells. Science, 2002. 296(5569): p. 927–31. https://doi.org/10.1126/science.1067631 PMID: 11988579
- Tapinos N., Ohnishi M., and Rambukkana A., ErbB2 receptor tyrosine kinase signaling mediates early demyelination induced by leprosy bacilli. Nature Medicine, 2006. 12(8): p. 961–966. <u>https://doi.org/10. 1038/nm1433</u> PMID: 16892039
- Constantin G., et al., Induction of adhesion molecules on human Schwann cells by proinflammatory cytokines, an immunofluorescence study. Journal of the Neurological Sciences, 1999. 170(2): p. 124– 130. PMID: 10561528

- Fujioka T., Kolson D.L., and Rostami A.M., Chemokines and peripheral nerve demyelination. Journal of NeuroVirology, 1999. 5(1): p. 27–31. PMID: 10190687
- **33.** Andrade P.R., et al., Inflammatory Cytokines Are Involved in Focal Demyelination in Leprosy Neuritis. J Neuropathol Exp Neurol, 2016.
- Shetty, V.P., Pathomechanisms of Nerve Damage, in IAL Textbook of Leprosy, H. Kumar and B. Kumar, Editors. 2010, Jaypee Brothers Medical Publishers (P) Ltd.: New Delhi.
- **35.** Khanolkar-Young S., et al., Tumor necrosis factor-alpha (TNF-α) synthesis is associated with the skin and peripheral nerve pathology of leprosy reversal reactions. Clinical and Experimental Immunology, 1995. 99(2): p. 196–202. PMID: 7851011
- Lockwood D.N. and Saunderson P.R., Nerve damage in leprosy: A continuing challenge to scientists, clinicians and service providers. International Health, 2012. 4(2): p. 77–85. https://doi.org/10.1016/j. inhe.2011.09.006 PMID: 24029146
- Ford A.L., Britton W.J., and Armati P.J., Schwann cells are able to present exogenous mycobacterial hsp70 to antigen-specific T lymphocytes. Journal of Neuroimmunology, 1993. 43(1–2): p. 151–160. PMID: 7681446
- Steinhoff U., et al., Mycobacterium leprae renders Schwann cells and mononuclear phagocytes susceptible or resistant to killer cells. Infection and Immunity, 1991. 59(2): p. 684–688. PMID: <u>1898912</u>
- Job C.K., Pathology and pathogenesis of leprous neuritis; a preventable and treatable complication. International Journal of Leprosy and Other Mycobacterial Diseases, 2001. 69(2 SUPPL.): p. S19–S29.
- 40. Burnett M.G. and Zager E.L., Pathophysiology of peripheral nerve injury: a brief review. Neurosurg Focus, 2004. 16(5): p. E1. PMID: 15174821
- Miko TL, Le Maitre C, Kinfu Y, Damage and regeneration of peripheral nerves in advanced treated leprosy. Lancet, 1993. 342:521–525 PMID: 8102668
- Becx-Bleumink M., Berhe D., and Mannetje W., The management of nerve damage in the leprosy control services. Lepr Rev, 1990. 61(1): p. 1–11. PMID: 2319899
- van Veen N.H., et al., Corticosteroids for treating nerve damage in leprosy. A Cochrane review. Lepr Rev, 2008. 79(4): p. 361–71. PMID: 19274983
- Croft R.P., et al., The treatment of acute nerve function impairment in leprosy: Results from a prospective cohort study in Bangladesh. Leprosy Review, 2000. 71(2): p. 154–168. PMID: 10920611
- Richardus J.H., et al., Treatment with corticosteroids of long-standing nerve function impairment in leprosy: a randomized controlled trial (TRIPOD 3). Lepr Rev, 2003. 74(4): p. 311–8. PMID: 14750576
- Smith W.C., et al., Steroid prophylaxis for prevention of nerve function impairment in leprosy: randomised placebo controlled trial (TRIPOD 1). BMJ, 2004. 328(7454): p. 1459. <u>https://doi.org/10.1136/ bmj.38107.645926.AE PMID: 15159285</u>
- Lambert S.M., et al., Comparison of Efficacy and Safety of Ciclosporin to Prednisolone in the Treatment of Erythema Nodosum Leprosum: Two Randomised, Double Blind, Controlled Pilot Studies in Ethiopia. PLoS Neglected Tropical Diseases, 2016. 10(2).
- Lambert S.M., et al., A Randomized Controlled Double Blind Trial of Ciclosporin versus Prednisolone in the Management of Leprosy Patients with New Type 1 Reaction, in Ethiopia. PLoS Negl Trop Dis, 2016. 10(4): p. e0004502. https://doi.org/10.1371/journal.pntd.0004502 PMID: 27046330
- van Veen N.H., et al., Decompressive surgery for treating nerve damage in leprosy. Cochrane Database Syst Rev, 2009(1): p. CD006983.
- Chandler D.J., et al., Household Costs of Leprosy Reactions (ENL) in Rural India. PLoS Negl Trop Dis, 2015. 9(1): e0003431 https://doi.org/10.1371/journal.pntd.0003431 PMID: 25590638
- Bowers B., et al., Responding to the challenge of leprosy-related disability and ultra-poverty. Leprosy Review, 2014. Sep; 85(3):141–8. PMID: 25509714
- Wagenaar I., et al., Diet-Related Risk Factors for Leprosy: A Case-Control Study. PLoS Negl Trop Dis, 2015. 9(5): e0003766. h
- Oxford Centre for Evidence-based Medicine. Levels of Evidence. 2009 [cited 2016 4th of April]; Available from: http://www.cebm.net/oxford-centre-evidence-based-medicine-levels-evidence-march-2009/.